

Name of journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: 29101

Manuscript Type: Original Article

CHANGES: changes in red

1. Title in bold
2. Postal code
3. Correspondence to: mentioned again, although it has been left as was because the example format that was sent to us is like that.
4. Biostatistics statement, etc, attached.
5. Abstract: titles in bold
6. Author and title
7. Reference numbers in format stated
8. Misspelling corrected
9. Figures of endoscopy included (we have more if necessary).
10. Comments included.
11. DOIs included
12. Google title search: no title that is the same.

FINAL CHANGES (1/10/16): changes in blue

1. Images are cited (page 12) and discussed (page 16).
2. Spelling mistakes have been corrected.
3. Paging added.
4. Role of Golimumab added in abstract: Aims, page 4.
5. Comma removed in last line of conclusions of abstract so that Goli's short-term safeness is understood.
6. Introduction shortened (page 6).
7. Heading in methods (page 8).
8. Years included (page 11).
9. Discussion: reduced and mistakes corrected.
10. Note added in table 1 to explain that 31 of 33 patients had received steroids.